Author's response to reviews

Title: Anti-Helicobacter pylori activity and immunostimulatory effect of extracts from Byrsonima crassa Nied. (Malpighiaceae)

Authors:

Cibele Bonacorsi (bonac@terra.com.br)
Maria Stella G Raddi (raddims@fcfar.unesp.br)
Iracilda Z Carlos (carlosiz@fcfar.unesp.br)
Miriam Sannomiya (miriamsa@posgrad.iq.unesp.br)
Wagner Vilegas (vilegasw@iq.unesp.br)

Version: 3 Date: 30 December 2008

Author's response to reviews:

December 30, 2008

Editor
BMC Complementary and Alternative Medicine

Dear Dr.:

Cibele Bonacorsi, Maria Stella G Raddi, Iracilda Z Carlos, Miriam Sannomiya and Wagner Vilegas are resubmitting the revised manuscript “Anti-Helicobacter pylori activity and immunostimulatory effect of extracts from Byrsonima crassa Nied. (Malpighiaceae)” to BMC Complementary and Alternative Medicine. The changes were done as requested: references was cited in numerical order in the text and listed in the bibliography in the order they are mentioned in the text. The authors give permission to the BioMed Central Editorial Production Team to make changes on the Figures size for the best final layout.

Sincerely

Maria Stella Gonçalves Raddi
Faculdade de Ciências Farmacêuticas de Araraquara
Rua Expedicionários do Brasil, 1621 Caixa Postal 503,
CEP 14801-902, Araraquara, SP, Brazil
Fone: +55 16 33016557
Fax: +55 16 3301-6559 fax
E-mail address: raddims@fcfar.unesp.br